

Dear referee,

Thank you very much for your comments on my manuscript.

According to your decision I did not perform any revision.

Sincerely yours,

Glsm zlem Elpek, MD, Professor of Pathology

Akdeniz University, Medical School, Dumlupinar Bulvari,

Department of Pathology,

07070- Antalya- Trkiye